Author's response to reviews

Title: Testing the effectiveness of a mindfulness-based intervention to reduce emotional distress in outpatients with diabetes (DiaMind): design of a randomized controlled trial.

Authors:

Jenny van Son (J.vanSon1@uvt.nl)
Ivan Nyklicek (I.Nyklicek@uvt.nl)
Victor JM Pop (V.J.M.Pop@uvt.nl)
François Pouwer (F.Pouwer@uvt.nl)

Version: 2 Date: 27 January 2011

Author's response to reviews: see over
Dear Editor,

We would like to thank you for your email of January the 21st, and for giving us the opportunity to submit a revised version of our study protocol entitled ‘Testing the effectiveness of a mindfulness-based intervention to reduce emotional distress in outpatients with diabetes (DiaMind): design of a randomized controlled trial.’ for publication in BMC Public Health.

In addition, we would like to thank you for the comment on our protocol. In reaction to this comment we would like to mention that we already had included information about informed consent in our manuscript. However, in the revised manuscript we added that it concerned ‘a written’ informed consent. Hopefully this addition is sufficient. We have highlighted the change with a yellow color.

We look forward to receiving your decision.

On behalf of all authors,

Yours sincerely,

Jenny van Son MSc.